Author’s response to reviews

Title: Cefuroxime, levofloxacin, esomeprazole, and bismuth as first-line Helicobacter pylori eradication therapy in patients allergic to penicillin

Authors:
Zhiqiang Song (szqinputh@163.com)
Wei Fu (weierhsc@qq.com)
Liya Zhou (zhouliya123456@163.com)

Version: 2 Date: 23 Feb 2019

Author’s response to reviews:

Reviewer 1: I have one comment the authors should include the adverse reaction rate in table 5, as it would be interesting to know the adverse event rate in other studies as this seems a bit high.

Response: The adverse event rates have been included in the Table 5.